Author’s response to reviews

Title: Neurodevelopmental profile of Fetal Alcohol Spectrum Disorder: A systematic review

Authors:

Shannon Lange (shannon.lange@camh.ca)
Joanne Rovet (joanne.rovet@sickkids.ca)
Jürgen Rehm (jtrehm@gmail.com)
Svetlana Popova (lana.popova@camh.ca)

Version: 1 Date: 02 Feb 2017

Author’s response to reviews:

Dear Ms. Clark,

As requested, the PROSPERO registration number and date of registration has been added to last line of the abstract and the PRISMA checklist has been submitted as supplementary material.

Sincerely,

Shannon Lange, MPH, PhD Candidate
Institute for Mental Health Policy Research, Centre for Addiction and Mental Health
Graduate Student, Institute of Medical Science, University of Toronto

Contact information:
Institute for Mental Health Policy Research
Centre for Addiction and Mental Health
33 Russell Street, Room # T401
Toronto ON, Canada M5S 2S1
Tel: (416) 535-8501 ext. 34512